\documentclass{revtex4}

\usepackage[english]{babel}
\usepackage[T1]{fontenc}
\usepackage[latin9]{inputenc}
\usepackage{mathpazo}

\begin{document}

% Refereed publications are sorted by their "received" date

% 2003
\nocite{Carpet-Schnetter2003a}  % 2003-10-07

% 2004
\nocite{Carpet-Baiotti2004a}    % 2004-10-12

% 2005
\nocite{Carpet-Zink2005a}       % 2005-02-11
\nocite{Carpet-Sperhake2005a}   % 2005-03-16
\nocite{Carpet-Diener2005a}     % 2005-12-21
\nocite{Carpet-Diener2005b}     % 2005-12-22

% 2006
\nocite{Carpet-Schnetter2006a}
\nocite{Carpet-Sopuerta2006a}
\nocite{Carpet-Schnetter2006b}
\nocite{Carpet-Loeffler2006a}
\nocite{Carpet-Sperhake2006a}
\nocite{Carpet-Dorband2006a}
\nocite{Carpet-Baiotti2006a}
\nocite{Carpet-Ott2006a}
\nocite{Carpet-Reisswig2006a}
\nocite{Carpet-Bruegmann2006a}
\nocite{Carpet-Zink2006a}
\nocite{Carpet-Ott2006b}
\nocite{Carpet-Baiotti2006b}
\nocite{Carept-Pazos2006a}
\nocite{Carpet-Szilagyi2006a}
\nocite{Carpet-Baker2006a}

% 2007
\nocite{Carpet-Herrmann2007a}
\nocite{Carpet-Giacomazzo2007a}
\nocite{Carpet-Marronetti2007a}
\nocite{Carpet-Herrmann2007b}
\nocite{Carpet-Koppitz2007a}
\nocite{Carpet-Campanelli2007a}
\nocite{Carpet-Gonzalez2007a}
\nocite{Carpet-Campanelli2007b}
\nocite{Carpet-Thornburg2007a}
\nocite{Carpet-Ajith2007a}
\nocite{Carpet-Vaishnav2007a}
\nocite{Carpet-Herrmann2007c}
\nocite{Carpet-Krishnan2007a}
\nocite{Carpet-Brown2007a}      % 2007-07-20
\nocite{Carpet-Bruegmann2007a}  % 2007-08-07
\nocite{Carpet-Rezzolla2007a}   % 2007-08-29
\nocite{Carpet-Lousto2007a}     % 2007-08-30
\nocite{Carpet-Pollney207a}     % 2007-09-06
\nocite{CarpetResult-Buonanno2007a} % 2007-09-25
\nocite{Carpet-Campanelli2007c}     % 2007-10-03
\nocite{Carpet-Ajith2007b}          % 2007-10-15
\nocite{Carpet-Rezzolla2007b}
\nocite{CarpetResult-Boyle2007a}
\nocite{Carpet-Berti2007a}
\nocite{Carpet-Zink2007a}
\nocite{Carpet-Shoemaker2007a}

% 2008
\nocite{CarpetResult-Benger2007a}



\bibliographystyle{apsrev-titles-manyauthors}
\bibliography{publications}

\end{document}
